Author's response to reviews

Title: Reversible atrial fibrillation secondary to a mega-oesophagus

Authors:

Tahwinder Upile (waseem_wk1@yahoo.co.uk)
Waseem Jerjes (waseem_wk1@yahoo.co.uk)
Colin Hopper (waseem_wk1@yahoo.co.uk)
Mohammed El Maaytah (waseem_wk1@yahoo.co.uk)
J Mahil (waseem_wk1@yahoo.co.uk)

Version: 2  Date: 1 December 2006

Author’s response to reviews:

Dear Sir/Madame

Response to Reviewer: Samuel Levy

We are grateful for the suggestions and have made all appropriate changes. We are grateful for you assessment of the paper and for its acceptance.

Changes point by point include:-
Abstract
Page 2 line 5 as suggested by reviewer 6% changed to 5%

Introduction
Page 5 line 4 suggested reviewer 6% changed to 5%

Discussion/Conclusions
As suggested the anatomical proximity of the oesophagus with the left atria is mentioned on page 6 line 6.
One reference added.

Response to Reviewer: Robert Roberts

We are grateful for you assessment of the paper and for its acceptance.

Please note, Mr. BT is the patient's husband and he granted the consent for the publication

Many thanks

The authors